Author's response to reviews

Title: Health impact assessment and short-term medical missions: a methods study to evaluate quality of care

Authors:

Jesse Maki (jesse_maki@hms.harvard.edu)
Munirih Qualls (munirih_qualls@hms.harvard.edu)
Benjamin White (benjamin.white@post.harvard.edu)
Sharon Kleefield (sharon_kleefield@hms.harvard.edu)
Robert Crone (robert_crone@hms.harvard.edu)

Version: 3 Date: 15 May 2008

Author's response to reviews: see over
Melissa Norton  
Editor in Chief, BMC Health Services Research  

May 15, 2008  

Dear Dr. Norton,  

Enclosed, please find the updated, reformatted manuscript entitled ‘Health impact assessment and short-term medical missions: a methods study to evaluate quality of care’ which we respectfully submit for consideration in *BMC Health Services Research*.  

Please note, I previously sent an email questioning whether or not the use of www.stmmconnect.org needs to be cited as a reference, because it is not a reference but rather part of our work. I have removed any use of the site where able and in 2 other areas I have used only stmmconnect.org without the www designation. I hope this is adequate. All websites that are sources are formatted and included in the references through endnote.  

I will serve as the corresponding author. We very much appreciate working with you and your continued consideration of this manuscript.  

Kind Regards,  
Jesse  

Jesse L. Maki  
Harvard Medical School